Author's response to reviews

Title: Single Ventricle with Persistent Truncus Arteriosus: A Case Report of Two Rare Entities in an Adult Patient

Authors:

Ina Porter (iporter@kumc.edu)
James L. Vacek (jlvacek@mac.md)

Version: 6 Date: 31 January 2008

Author's response to reviews: see over
January 25, 2008

JMCR Editorial Team

Re: MS: 1651213828132184 - Single Ventricle with Persistent Truncus Arteriosus: A Case Report of Two Rare Entities in an Adult Patient

Dear Editorial Team:

Attached is our re-revised version of the above manuscript. We have completely and successfully addressed all the concerns of the referees.

One referee was completely satisfied with our prior revision, had no additional comments and recommended accept.

The 2nd referee had two suggestions:
1. Discuss the patient’s balanced circulation which promoted survival. We have added extensive additional discussion in the first paragraph of page 10 regarding this.

2. Discuss points of management to be derived from this patient’s outcome. We have added extensive discussion of present-day approaches to management in paragraph 2 of page 10 and paragraph 1 of page 11.

Essentially 3 paragraphs of new information and discussion have been added, as well as 3 new references (#’s 20-22.)

The 2nd referee’s comments have been completely addressed as appropriate for a case report such as this.

This referee also stated that he could not view the attached ECG (Figure 1). We have submitted it with each version of our manuscript, and attach it again.

We sincerely hope that you will find our re-revised manuscript suitable for publication.

James L. Vacek, MD, FACC, FACP, FAHA
Director of Cardiovascular Research
The Center for Cardiovascular Scholarship
Mid America Cardiology
The University of Kansas Hospital
Kansas City, KS  66160
913-588-9655